RICH1 phone meeting 25/05/07

People present:

Bristol:  Richard.

CERN: Carmelo, Dave, Fabio (part), Kenneth, Olav, Richard, Tim (part).

Imperial: Bill.

Oxford: Neville (part), Rohan.

1. Beam Pipe Commissioning.

There have been delays in pumping down, and this will not be done until Tuesday the 28th.  This is not expected to delay the overall commissioning schedule.  Commissioning is due to finish on the 24th June.

2. Spherical mirrors.

The alignment set up is ready, with the bench and rig in place.  Alignment is due to begin at the end of LHCb week (the week starting the 4th June) or the beginning of the week after.

The samples from CMA, which were exposed to C4F10, have been measured once more.  No change in the reflectivity, radius of curvature or D0, surface roughness or mechanical properties where observed in any of the samples.  The small mirror and the uncoated sample have had 11 months exposure, the larger demonstrator mirror has had 10.  The mirrors are now in gas once more and will be re-measured in about five months.

3.  Flat mirrors.

The upper and lower plates have been surveyed and adjusted into the correct position (since the time of the last meeting it has been found that a mistake was made in this procedure, so there is a need for the angle of the plates to be readjusted.  Previously the v-blocks where not lowered onto the support bar, but were still supported by the Teflon tipped screws).  The 8 lower mirrors have been put on the two plates and these installed on the alignment rig.  One mirror on one plate is aligned with the plate using calipers and the other three aligned with respect to this one using the reflection of a poster of straight lines.  One of the mirrors on the other plate is then aligned with these and used as a reference to align the others.

Some mechanical alterations have been needed.  A thread insert within the plate needed to be replaced and has been.  Extra holes in the v-block of the upper mirror plates have been added.  It is possible that further holes in the back of the rig support plate will be needed to aid alignment and adjustment.  This will depend on how well alignment can be done using the procedure described above. 

4. HPD Mounting Assembly.

4.1. Upper Assembly.

The safety review has been completed and no objections were made.  The process for the documentation approval is somewhat unclear however.  The following comments were made at the review:

There is the possibility that rather than taking the HPD assembly over the detector to the c-side, that it can be taken down the PM lift and taken through the cryogenic area on a trolley.  For this to be possible a trolley no more than 80cm across is needed.

It would be preferable for the assembly to be delivered to CERN in pieces.

A test installation of the internal pieces and external rails can be made after deliveries from Oxford have been made.

A procedure needs to be defined to ensure HV is turned off during maintenance operation (action Gabriel)

Fabrication is still ongoing at NAB, but it is promised by the 15th of June.  This is a delay with respect to previous information.  It is needed in CERN for the end of June, with tests to be done in Oxford first.

4.2.  Lower Assembly.

A new model is being worked on by Brian.  This is going well and drawings should be sent to Imperial at the end of week beginning 28th May.

The lower HPD assembly HPDs point upwards, meaning access has to be done from underneath, where there is no room.  This can be overcome by rotating the module about a horizontal axis to stand vertically, which allows better access for extraction.  To do this requires a significant framework.  It is now planned that wheels and lifting lugs will be added to this frame to allow it to be used to move the modules into position.  This will mean that ladders are to be assembled into the lower box in a vertical positions which will necessitate minor changes to the support rods.

4.3. Column dressing.

The only change since the last meeting was that the week starting 21st May was spent cabling, rather than on dressing the 7 columns. Cooling plates are to be checked the week beginning 28th May and the final assembly 4th June.  The mechanics for the ladder are complete and at CERN.  We are still awaiting delivery of cups, which should arrive 40 days after the order which was placed, roughly two weeks prior to the meeting.  The spare columns will not be finished in June due to the cup situation.

4.4 HPD Mechanics.

Manufacture of the upper chariot is in progress in the IC workshop.  In the meantime the hood and manifold are being designed.

5. Cabling.

All of the wire cables for the upper box have been done.  All that remains is the optical fibres, LAMS and MDMS cables from the upper tray.  The LV power cables have been put in to the bottom shelf.  An area on the lower level floor has been marked to be cut out to allow for access for cables.  Light cables for the lower part are to be done the week beginning the 28th.

6.  Light leak detector.

Two PMT-based devices of similar dimensions that could be used have been found.  A decision on which one to use will need to consider the suitability of the cabling, as the two devices require different numbers of cables.

7. MDMS.

There has been no MDMS meeting since the last RICH1 phone meeting. After the meeting Ray reported the following:

The stages and power supply/controller have been ordered

A mock-up of the photon funnel is in the Syracuse workshop

Prototyping of collimators (pin-hole pairs) and of the electronics logic for the LEDs is in progress.  A new schedule should be worked out, taking into consideration the LHC delay and the desirability to test in the SSB2 first.

8. LAMS

It has been agreed that this should be installed as soon as possible, although it is not known if the parts have yet been ordered.

9. Quartz window.

The schedule for glueing the quartz windows and the installation of the photon funnel needs to be reconsidered in the light of the LHC delay and the pre-testing of the MDMS in the SSB2 (action Dave)
